

JGR Oceans

RESEARCH ARTICLE

10.1029/2022JC019061

Key Points:

- The average rate of sea-level rise in the Mediterranean Sea increased from -0.3 mm yr⁻¹ in the period 1960-1989 to 3.6 mm yr⁻¹ in 2000-2018
- Sterodynamic and land-mass changes both have made comparable contributions to the Mediterranean sea-level rise since 2000
- Since 2000, sea level has been rising faster in the Adriatic, Aegean, and Levantine Seas than anywhere else in the Mediterranean Sea

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: F. M. Calafat, Francisco.calafat@noc.ac.uk

Citation:

Calafat, F. M., Frederikse, T., & Horsburgh, K. (2022). The sources of sea-level changes in the Mediterranean Sea since 1960. *Journal of Geophysical Research: Oceans*, *127*, e2022JC019061. https://doi.org/10.1029/2022JC019061

Received 5 JUL 2022 Accepted 12 SEP 2022

© 2022 Jet Propulsion Laboratory and National Oceanography Centre. California Institute of Technology. Government sponsorship acknowledged. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

The Sources of Sea-Level Changes in the Mediterranean Sea Since 1960

F. M. Calafat¹, T. Frederikse², and K. Horsburgh^{1,3}

¹National Oceanography Centre, Liverpool, UK, ²Jet Propulsion Laboratory, California Institute of Technology, Pasadena, CA, USA, ³Green Climate Fund, Yeonsu-gu, Republic of Korea

Abstract Past sea-level changes in the Mediterranean Sea are highly non-uniform and can deviate significantly from both the global average sea-level rise and changes in the nearby Atlantic. Understanding the causes of this spatial non-uniformity is crucial to the success of coastal adaptation strategies. This, however, remains a challenge owing to the lack of long sea-level records in the Mediterranean. Previous studies have addressed this challenge by reconstructing past sea levels through objective analysis of sea-level observations. Such reconstructions have enabled significant progress toward quantifying sea-level changes, however, they have difficulty capturing long-term changes and provide little insight into the causes of the changes. Here, we combine data from tide gauges and altimetry with sea-level fingerprints of contemporary land-mass changes using spatial Bayesian methods to estimate the sources of sea-level changes in the Mediterranean Sea since 1960. We find that, between 1960 and 1989, sea level in the Mediterranean fell at an average rate of -0.3 ± 0.5 mm yr⁻¹, due to an increase in atmospheric pressure over the basin and opposing sterodynamic and land-mass contributions. After 1989, Mediterranean sea level started accelerating rapidly, driven by both sterodynamic changes and land-ice loss, reaching an average rate of 3.6 ± 0.3 mm yr⁻¹ in the period 2000–2018. The rate of sea-level rise shows considerable spatial variation in the Mediterranean Sea, primarily reflecting changes in the large-scale circulation of the basin. Since 2000, sea level has been rising faster in the Adriatic, Aegean, and Levantine Seas than anywhere else in the Mediterranean Sea.

Plain Language Summary The rate of sea-level rise varies considerably across the Mediterranean basin. Characterizing this geographic variability is essential for the design of coastal adaptation plans. This, however, is no easy feat because of the sparseness of the observational sea-level record. In the Mediterranean Sea, long tide gauge records are restricted to a few locations along the European coastlines. Satellite altimetry provides sea-level observations with better spatial coverage but only since 1992. Past studies have addressed the difficulties related to data sparseness by combining data from tide gauges and altimetry to produce sea-level reconstructions. However, existing sea-level reconstructions have difficulty capturing long-term changes and do not distinguish between the causes of the changes. Here, we analyze sea-level observations in combination with fingerprints of land-ice melting using probabilistic methods to show that the average rate of sea-level rise in the Mediterranean Sea increased from -0.3 mm yr^{-1} in the period 1960–1989 to 3.6 mm yr⁻¹ in 2000–2018, as a result of increased ocean warming and land-ice melting. Furthermore, we find that, since 2000, sea-level in the Adriatic, Aegean, and Levantine Seas has been rising faster than anywhere else in the Mediterranean Sea, due to changes in the circulation of the basin.

1. Introduction

Global mean sea level has been rising more than two times faster in the past three decades (\sim 3.3 mm yr⁻¹) than throughout most of the twentieth century (Dangendorf et al., 2017; Frederikse et al., 2020; Hay et al., 2015). This acceleration has been largely driven by increased rates of both land-ice melting and ocean thermal expansion (Dangendorf et al., 2019; Frederikse et al., 2020). Both of these processes give rise to highly non-uniform patterns of sea-level change, implying that the local rate of sea-level change can deviate significantly from the global average rate. In the case of land-ice melting, the associated patterns of (static-equilibrium) sea-level change are unique to each melt source (Clark & Lingle, 1977) and are known as sea-level fingerprints. This spatial non-uniformity in the sources of global sea-level rise highlights the need for accurate knowledge of how sea level changes at a local and regional scale.

The Mediterranean Sea is a perfect illustration of why regional assessments of sea-level rise are crucial to coastal adaptation planning. For one thing, the Mediterranean is among the world's most vulnerable regions to climate change, with 86% of the region's World Heritage sites already at risk from coastal flooding and erosion (Reimann et al., 2018). For another, past sea-level changes in the Mediterranean Sea have been found to exhibit considerable spatial variation as well as to deviate significantly from both the nearby Atlantic and the global average (Calafat & Gomis, 2009; Marcos & Tsimplis, 2008; Orlić et al., 2019; Tsimplis & Baker, 2000). In particular, there is observational evidence that, between 1960 and 2000, sea level in the Mediterranean remained flat or slightly fell (depending on the study), in contrast to the Eastern North Atlantic where sea level rose at a rate of more than 1 mm yr⁻¹ (Marcos & Tsimplis, 2008; Tsimplis et al., 2005; Tsimplis & Baker, 2000). The small rates of sea-level change in the Mediterranean over 1960-2000 were partly due to a persistent increase in atmospheric pressure over the basin related to the North Atlantic Oscillation (NAO) (Gomis et al., 2008; Tsimplis et al., 2005; Tsimplis & Josey, 2001). Since the early 1990s, the Mediterranean has seen a significant increase in the rate of sea-level rise, leading to an average rate of 2.7 mm yr⁻¹ for the period 1993–2017 (Mohamed et al., 2019), which is on a par with the global average rate. The relative contributions from sterodynamic changes (i.e., changes in ocean density and circulation) and land-ice melting to this recent increase in the rate of Mediterranean sea-level rise remain unclear. Characterizing these spatiotemporal sea-level variations in the Mediterranean Sea and understanding their causes is a priority, in order to enable more effective coastal adaptation. However, achieving this goal is challenging because of the sparseness of the observational sea-level record in both time and space.

Tide gauge records are our main source of information on long-term sea level changes, but long records are restricted to a few coastal locations. In the Mediterranean Sea, the tide gauge network is heavily biased toward the European shorelines, with only one relatively long record (~35 years) available on the African coasts. Satellite altimetry provides observations with better spatial coverage but only since 1992. Past studies have attempted to address the difficulties related to data sparseness by statistically combining tide gauge records with altimetry data to produce sea-level reconstructions. This is typically done by computing empirical orthogonal functions (EOFs) from the altimetry data (or sometimes from modeled data) and then fitting a subset of those to the tide gauge data to estimate the temporal amplitude associated with each retained EOF (Calafat et al., 2014; Calafat & Gomis, 2009; Calafat & Jordà, 2011; Meyssignac et al., 2011). This approach yields sea-level fields with the same spatial coverage as the altimetry data and spanning the same period as the tide gauge records. These fields provide valuable insights into the spatial structure of long-term sea-level changes and can also be used to compute spatially averaged changes. However, EOF-based reconstructions have difficulty capturing the long-term trends (Calafat et al., 2014), do not adequately account for differences between relative sea level (RSL, as observed by tide gauges) and geocentric sea level (GSL, as observed by altimetry), and do not have the ability to separate the causes of sea-level changes.

Here, we combine data from tide gauges and altimetry with sea-level fingerprints of contemporary land-mass changes using a Bayesian hierarchical model (BHM) (see Banerjee et al., 2004 for a general description of hierarchical models) to reconstruct sea-level changes in the Mediterranean Sea since 1960 and quantify the source contributions to such changes. Our approach has three key advantages over EOF-based reconstructions: (a) it explicitly distinguishes between RSL and GSL; (b) it separates long-term sea-level changes from short-term variability as well as large-scale signals from small-scale changes; and (c) it quantifies the individual contributions from sterodynamic changes, land-mass changes, and glacial isostatic adjustment (GIA, Tamisea, 2011) to long-term sea-level changes. The investigation of the new Bayesian solutions allows us to answer fundamental questions about sea level in the Mediterranean Sea, such as how much sea level in the basin has risen since 1960, whether the rate of change has been constant, and the origin of spatial variation in the sea-level changes.

2. Data and Methods

2.1. Observational Data and Sea-Level Fingerprints

Monthly mean values of sea level from tide gauge records are obtained from the Revised Local Reference data archive of the Permanent Service for Mean Sea Level (PSMSL, https://www.psmsl.org/) (Holgate et al., 2013). Monthly values flagged as suspicious by the PSMSL are rejected. The inverse barometer (IB) effect is removed from the monthly time series based on sea-level pressure data from the NCEP/NCAR reanalysis (Kalnay et al., 1996). Annual and semi-annual cycles are also removed through ordinary least squares. The adjusted monthly data are then converted into annual time series for our analysis, noting that only years with at least 10





Figure 1. Location of the tide gauge stations used in the Bayesian reconstruction (black and colored dots). The colored dots denote the location of sites also used in the evaluation of the Bayesian hierarchical model (BHM) (Marseille, Trieste, Khalkis South, and Alexandria). The shading indicates regions into which we split the Mediterranean basin for the purpose of computing regionally averaged sea-level rates.

valid monthly values are considered. The final tide gauge data set comprises 2292 annual values from a total of 107 tide gauge records spanning the period 1960–2018. The location of the tide gauges is shown in Figure 1.

The satellite altimetry data are obtained from the multi-mission gridded sea surface height product provided by the Copernicus Marine Environment Monitoring Service (available at http://marine.copernicus.eu/). The data are provided as weekly fields on a $1/4^{\circ} \times 1/4^{\circ}$ near global grid covering the period from January 1993 to December 2018. These weekly fields are averaged into annual fields for our analysis. The data are made available with all standard corrections applied, including corrections for tropospheric (wet and dry) and ionospheric path delays, sea state bias, tides (solid earth, ocean, loading, and pole), and atmospheric effects as quantified by the Mog2D barotropic model (Carrère & Lyard, 2003) for periods <20 days and the IB effect for longer periods.

The sea-level fingerprints associated with contemporary land-mass changes as well as the GIA estimates are the same as derived by Frederikse et al. (2020). The sea-level fingerprints represent the gravitational, rotational, and deformation (GRD) effects on sea level due to contemporary mass changes of

the Greenland and Antarctic ice sheets, glaciers, and terrestrial water storage. The fingerprints are provided as an ensemble (5000 members) of annual sea-level fields for the period 1960–2018. Because GRD effects in our BHM are described in terms of rates rather than levels (to facilitate separation from the short-term variability), we convert the fingerprints from sea levels to yearly rates of change. This is done by first smoothing the sea-level time series at each grid point using an integrated-random-walk-plus-noise model that we fit through a Kalman smoother and then by taking first differences of the smoothed time series. This procedure removes any short-term variability that might exist in the original time series (see Figure S1 in Supporting Information S1 for an example). The GIA data are provided as an ensemble (5000 members) of time-invariant patterns of sea-level change rates. The ensembles allow us to derive probability distributions for the land-mass and GIA contributions, which we use as prior distributions in the BHM as described later. We note that both the sea-level fingerprints and the GIA patterns are different for RSL and GSL, and we take account of this when building the BHM.

2.2. BHM of Sea Level

Our goal here is to reconstruct sea-level changes in the Mediterranean Sea for the period 1960–2018 at any arbitrary location and quantify the individual source contributions to such changes. We achieve this by analyzing observations from tide gauges and satellite altimetry using a spatiotemporal probabilistic model. We adopt a Bayesian hierarchical approach, meaning that we use probability distributions to describe uncertainty in all model quantities (i.e., observations, source contributions, and unknown parameters). In designing the model, it is important to recognize that while tide gauges measure sea level relative to the land on which they reside (i.e., RSL), satellite altimetry measures sea level relative to the Earth's center of mass (i.e., GSL). This means that tide gauge measurements reflect changes in both sea surface height (SSH) and the underlying solid Earth (i.e., vertical land movements, VLMs), while altimeters only sense changes in SSH. Our model rigorously accommodates these differences between tide gauges and altimetry.

Another point worth noting is that the sea-level observations are subject to significant uncertainty arising from measurement error, uncertain altimetry corrections, etc. To account for this uncertainty, we assume that the observations are noisy measurements of a latent process which represents the true changes in sea level. The true latent sea-level process is then modeled as the sum of five contributions (or processes), namely sterodynamic changes (x_{SD}), contemporary GRD effects (x_{GRD}), GIA (x_{GIA}), small-scale changes (x_S), and short-term variability (x_V). The process x_S captures long-term trends due to small-scale processes, including VLM from tectonics, sediment compaction, and groundwater. The model is specified as a BHM with three levels: (a) a probability model that describes the distribution of the sea-level observations conditional on the true latent process (data model); (b) a probability model that describes the dynamics of the five source contributions conditional on a set of parameters (process model); and (c) a prior distribution that describes the uncertainty in the model parameters and encodes our prior knowledge about the data and the processes (parameter model). Inferences in our model



are made by sampling from the posterior distribution of the processes and parameters given the observations, which is proportional to the product of the three probability models that form the hierarchy. We note that the BHM assimilates sea-level data corrected for the IB effect, and thus it does not estimate the contribution from changes in atmospheric pressure. Such contribution is quantified separately based on data from the NCEP/NCAR reanalysis and assuming an IB effect. Next, we describe the three layers of the BHM.

Let $S_i = \{s_{i,1}, \dots, s_{i,N_i}\}, i \in \{RSL, GSL\}$, denote the set of measurement sites (tide gauges for i = RSL and altimetry for i = GSL). Further, denote $z_{i,t}(s_i)$ as the sea level observation at time step *t* and location s_i , for $s_i \in S_i$ and $i \in \{RSL, GSL\}$. Then, the data level of the BHM can be written as

$$z_{i,t}(\mathbf{s}_i) = y_{i,t}(\mathbf{s}_i) + b_i(\mathbf{s}_i) + e_{i,t}(\mathbf{s}_i)$$
(1)

where $y_{i,t}(\mathbf{s}_i)$ denote the true latent sea level (RSL or GSL), $b_i(\mathbf{s}_i)$ represent data offsets which are modeled as i.i.d. normal random variables $b_i(\mathbf{s}_i) \stackrel{iid}{\sim} N(0, \sigma_b^2)$, and the $e_{i,t}(\mathbf{s}_i)$ are assumed to be serially and spatially uncorrelated observation errors $e_{i,t}(\mathbf{s}_i) \stackrel{ind}{\sim} N(0, \sigma_b^2)$. Equation 1 is evaluated for t = 1, ..., T when i = RSL but only for t = 34, ..., T when i = GSL since the altimetry record starts in 1993 (this is also the case for any of the time-varying processes described below whenever these are subscripted by i). The altimetry data are subsampled at every five grid points in order to reduce the computational cost of the BHM.

Next, we describe the process level of the BHM. Except for the sea-level variability, all processes are described in terms of rates of sea-level change, not sea levels. This facilitates the separation of the variability from the long-term processes since it constrains the processes associated with long-term changes to be smooth. We assume that there is no difference between RSL and GSL in what refers to sterodynamic changes and short-term variability, whereas the contributions from GRD effects, GIA, and small-scale changes may be different for RSL and GSL. We recognize that sterodynamic changes can induce GRD effects through an influence on ocean bottom pressure, but such effects are generally small (Richter et al., 2013) and thus we ignore them.

The instantaneous rate of change due to sterodynamic changes (x_{SD}) is modeled as the sum of a spatiotemporal process that captures the effect of sterodynamic changes inside the Mediterranean (*I*) plus a spatially uniform time-varying process that captures the effect of changes outside the basin (*U*). Note that both processes (*I* and *U*) can affect the basin-average rate of change. The use of a spatially uniform process (*U*) is motivated by the physical expectation that, on long time scales, any significant difference in sea level between the Mediterranean and the nearby Atlantic will tend to be rapidly compensated via changes in the Atlantic inflow through the Strait of Gibraltar (Calafat, Chambers, & Tsimplis, 2012). This compensation takes the form of a spatially uniform barotropic signal. The process *I* is assumed to evolve according to a spatiotemporal random walk:

$$I_t(\mathbf{s}) = I_{t-1}(\mathbf{s}) + \omega_t(\mathbf{s})$$
(2)

where $\omega_t(\mathbf{s})$ is a zero-mean Gaussian process (GP) (Rasmussen & Williams, 2006) $\omega_t(\mathbf{s}) \sim \text{GP}(0, \mathbf{c}(\mathbf{s}, \mathbf{s}'; \sigma_I, \rho_I))$, with $\mathbf{c}(\cdot, \cdot)$ being a covariance function and σ_I and ρ_I denoting, respectively, the standard deviation and length scale that define the covariance function. The initial state of $I(I_0)$ is modeled as $I_0(\mathbf{s}) \sim \text{GP}(0, \mathbf{c}(\mathbf{s}, \mathbf{s}'; \sigma_{I_0}, \rho_{I_0}))$. For the covariance function of the GPs we assume a Matérn kernel with smoothing parameter v = 5/2.

The spatially uniform process, U_t , is assumed to follow a univariate random walk:

$$U_t = U_{t-1} + v_t \tag{3}$$

where $v_t \sim N(0, \sigma_U^2)$. The initial value, U_0 , is modeled in the parameter layer of the BHM by placing a prior distribution on it (in practice, Equation 3 is evaluated using backward recursion, and thus $U_0 := U_{t=T}$).

With this, the sterodynamic contribution to the rate of sea-level change is $x_{SD,t}(\mathbf{s}) = I_t(\mathbf{s}) + U_t$.

The rate of sea-level change due to contemporary GRD effects is modeled as:

$$x_{GRD_{i},t}(\mathbf{s}) = \mu_{GRD_{i},t}(\mathbf{s}) + \delta_{i,t}(\mathbf{s})$$
(4)

$$\delta_{i,t}(\mathbf{s}) = \delta_{i,t-1}(\mathbf{s}) + r_{i,t}(\mathbf{s})$$
(5)

$$\boldsymbol{r}_{i,t}(\mathbf{s}) \sim \mathrm{GP}\left(\boldsymbol{0}, \boldsymbol{\Sigma}_{\delta_i}\right) \tag{6}$$



where $i \in \{RSL, GSL\}$, $\mu_{GRD_i,i}(\mathbf{s})$ is the ensemble-mean fingerprint of the rate of sea-level change associated with GRD effects from Frederikse et al. (2020) and $\delta_{i,i}(\mathbf{s})$ is an error term that accounts for uncertainty in the fingerprint and is modeled as a spatiotemporal random walk. The initial value of δ_i is modeled as $\delta_{i,0}(\mathbf{s}) \sim GP(0, \Sigma_{\delta_{i,0}})$. The covariance matrices Σ_{δ_i} and $\Sigma_{\delta_{i,0}}$ are extracted from the ensemble of RSL and GSL fingerprints (i.e., they capture variation across the ensemble members after removal of the ensemble mean).

The contribution from GIA is modeled as a stationary Gaussian process:

$$x_{GIA_i}(\mathbf{s}) \sim \mathrm{GP}(\mu_{GIA_i}(\mathbf{s}), \Sigma_{GIA_i}) \tag{7}$$

where $i \in \{RSL, GSL\}$, and $\mu_{GIA_i}(s)$ and Σ_{GIA_i} are, respectively, the mean and covariance matrix of the ensemble of GIA rates from Frederickse et al. (2020).

The rate of sea-level change due to small-scale processes is modeled as i.i.d. normal random variables (i.e., time-invariant trends):

$$x_{S_i}(\mathbf{s}) \stackrel{iid}{\sim} N\left(0, \sigma_{S_i}^2\right)$$
 (8)

where $i \in \{RSL, GSL\}$. Note that Equation 8 assumes the small-scale trends to be spatially uncorrelated. We tested another version of the BHM in which x_{S_i} was allowed to vary in time as VLMs are not always well described by a linear trend, but such version of the model provided a worse fit to the data according to convergence diagnostics (e.g., larger *R*-hat values) and the solutions were very similar to those presented in this paper.

The sea-level rates due to sterodynamic changes, GRD effects, GIA, and small-scale processes can be integrated and combined to estimate the total long-term sea-level changes:

$$h_{i,i}(\mathbf{s}) = h_{i,t-1}(\mathbf{s}) + x_{SD,t-1}(\mathbf{s}) + x_{GRD_i,t-1}(\mathbf{s}) + x_{GIA_i}(\mathbf{s}) + x_{S_i,t-1}(\mathbf{s})$$
(9)

where the initial value of h_i is set to zero, $h_{i,0}(\mathbf{s}) = 0$.

Finally, the short-term sea-level variability, $x_{V,t}$, is modeled in terms of EOFs:

$$x_{V,t}(\mathbf{s}) = \sum_{n=1}^{5} a_{n,t} EOF_n(\mathbf{s})$$
(10)

$$a_{n,t} \stackrel{ina}{\sim} N(0,\sigma_a^2) \tag{11}$$

where $a_{n,t}$ is the temporal amplitude associated with the *n*th leading EOF. The EOFs are determined a priori from the full-resolution altimetry data after removal of a quadratic polynomial fit, whereas the temporal amplitudes are unknown and need to be estimated from the observations as part of the BHM. The first two EOFs alone capture 50.6% of the variance in the detrended altimetry data, while the first five EOFs explain 69.1% of the variance. Beyond five EOFs, the fraction of variance explained increases slowly with each subsequent EOF, indicating that such higher order EOFs may not be statistically significant. Indeed, past studies have found that there is no upside to including more than five EOFs in Mediterranean sea-level reconstructions (e.g., Calafat & Jordà, 2011). Based on this, we decide to use only the five leading EOFs. The spatial structure of the first three EOFs along with their temporal amplitudes are shown in Figure S2 (in Supporting Information S1).

We have tested the sensitivity of our results to the number of retained EOFs by comparing Bayesian solutions based on 2, 5, and 10 EOFs. We find that estimates of RSL rates are consistent across the three models within one posterior standard deviation (see Table S1 in Supporting Information S1). However, we also note that when using 10 EOFs convergence of the Markov chains (i.e., posterior samples) is worse than in the other two cases, indicating that including more than a few EOFs degrades the performance of the BHM. It is also important to recognize that using EOFs is not the only way to describe the variability. In particular, during the design phase of the BHM, we considered an alternative approach in which the variability was modeled as a spatial first-order autoregressive (AR1) process (similar to Piecuch et al., 2017). We found that when using EOFs the BHM captures a larger fraction of the variability and provides a more adequate fit to the data (according to convergence diagnostics) than when using an AR1 process.

With all the contributions now defined, the true latent sea-level containing all contributions can be written as:

$$y_{i,t}(\mathbf{s}) = h_{i,t}(\mathbf{s}) + x_{V,t}(\mathbf{s})$$
(12)



Posterior Distribution Mean and 5%–95% Credible Interval for the Parameters of the Bayesian Hierarchical Model, Along With the Prior Distribution Ascribed to Each Parameter

Parameter	Units	Description	Mean	5th	95th	Prior distribution	<i>R</i> -hat	$n_{\rm eff}/it$
σ_{RSL}	cm	Observation error SD	1.93	1.88	1.98	half $-N(0,10^2)$	1.00	1.56
σ_{GSL}	cm	Observation error SD	1.36	1.33	1.38	half – $N(0, 10^2)$	1.00	1.80
σ_{I_0}	mm yr ⁻¹	GP SD (I_0)	0.79	0.40	1.27	half - N(0, 10)	1.00	0.47
ρ_{I_0}	degree	GP length scale (I_0)	3.45	1.85	5.99	invGamma(5, 20)	1.00	0.19
U_0	$mm yr^{-1}$	Initial state $(U_{t=T})$	1.53	0.19	2.88	N(0, 10)	1.00	1.19
$\sigma_{S_{RSL}}$	$mm yr^{-1}$	SD (small-scale)	1.68	1.41	1.99	half - N(0, 10)	1.00	0.64
$\sigma_{S_{GSL}}$	$mm yr^{-1}$	SD (small-scale)	0.52	0.40	0.62	$half\!-\!N(0,2)$	1.01	0.12
σ_a	-	SD (EOF amplitude)	1.11	1.01	1.21	half $- N(0, 2)$	1.00	0.54

Note. The potential scale reduction statistic (*R*-hat) and the effective sample size per iteration (n_{eff}/it) are also shown. In general, *R*-hat should be close to 1 at convergence, whereas $n_{\text{eff}}/it > 0.003$ indicates low autocorrelation.

The BHM is completed by specifying the parameter model. Most of the unknown parameters are given prior distributions (i.e., they are modeled), while a few parameters are set to a fixed value based on prior knowledge or domain expertise. The prior distributions ascribed to the modeled parameters as well as the posterior mean and credible interval for such parameters are summarized in Table 1. The fixed parameters include σ_b , σ_I , σ_U , and ρ_I . The standard deviation σ_b is set equal to 1 m, which implies a diffuse prior distribution on the data offsets b_i (s_i) (the time series from both tide gauges and altimetry have been processed in such a way that offsets are <1 m). Both σ_I and σ_U are set equal to 0.3 mm yr⁻², which encapsulates our prior belief that the sterodynamic rate is unlikely to change by significantly more than 0.3 mm yr⁻¹ between consecutive years. Finally, the length scale ρ_I is set equal to 5°, which is necessary to ensure that I_I reflects large-scale changes, otherwise the model will favor smaller length scales owing to the relatively high resolution of the altimetry data for the period 1993–2018 (small length scales are not resolvable prior to 1993 due to the sparseness of the tide gauge data).

To make the problem computationally tractable, we approximate all GPs using nearest neighbor Gaussian processes with 20 neighbors, following the formulation given by Datta et al. (2016). The BHM is fitted using the No-U-Turn sampler (a variant of Hamiltonian Monte Carlo with adaptive optimization) as implemented by the Stan probabilistic programming language (Carpenter et al., 2017). We run the sampler with four chains of 1250 iterations each (warm-up = 500) for a total of 3000 post-warm-up draws.

The BHM provides estimates of sea-level changes, the yearly rate of sea-level change, and the contribution from each source for the period 1960–2018, both at tide gauge sites and on a $1/4^{\circ} \times 1/4^{\circ}$ grid. Because our estimates take the form of samples from the joint posterior distribution of the processes at all locations and times, we are able to compute spatial and temporal averages of any quantities of interest with quantifiable and meaningful uncertainty. In the following, our estimates are summarized by posterior means and standard deviations (SDs).

As mentioned above, the contribution from atmospheric pressure changes is computed outside the BHM assuming an IB effect. The rates of change associated with such contribution are computed by fitting an integrated-random-walk-plus-noise model separately at each grid point via Stan.

3. Model Evaluation

We evaluate the BHM in the light of two questions: (a) does the sampler accurately characterize the posterior distribution?; and (b) does the BHM provide an adequate fit to the data? The first question is addressed through a number of Markov chain Monte Carlo (MCMC) diagnostics that aim to diagnose problems with the sampler and assess convergence and mixing. Such diagnostics include the potential scale reduction statistic (\hat{R}), the effective sample size per iteration (n_{eff}/it), divergent transitions, and tree-depth saturation. We find that $\hat{R} < 1.05$ for all parameters in our model (\hat{R} should be close to 1 at convergence), suggesting that the Markov chains have converged to the equilibrium distribution and are providing a good approximation to the posterior distribution. We estimate n_{eff}/it to be >0.1 for all scalar parameters, indicating low autocorrelation and good mixing (The

Table 1

lower the autocorrelation the smaller the MCMC standard error). There were no divergent transitions during sampling and none of the iterations saturated the maximum tree depth, indicating that the sampler is able to fully explore the posterior distribution. The values of \hat{R} and n_{eff}/it for the scalar parameters of the BHM are summarized in Table 1.

To assess the adequacy and skill of the BHM, we begin by examining the model residuals at tide gauge sites (i.e., $Z_{RSLI} - v_{RSLI} - b_{RSL}$). The measurement model (Equation 1) assumes that the residuals are normally distributed, and so evaluating the validity of this assumption is an important self-consistency check. To this end, we test the null hypothesis that the residuals come from a normal distribution using the one-sample Kolmogorov-Smirnov test. The test returns a p-value = 0.4, thus failing to reject the null hypothesis and, in turn, indicating that the residuals conform to the normality assumption. We also note that the time-mean of the residuals across the tide gauges is distributed symmetrically around zero, with a mean value of 0.002 mm and a standard deviation of 0.08 mm, indicating no systematic deviations between our estimates and the observations. The excellent skill of the BHM in capturing the general structure of the data can be appreciated more clearly by plotting estimates of RSL against tide gauge records at a selection of four stations (Figure 2). The long-term changes are very well captured by the BHM at all sites, even though such changes can differ significantly across sites due to spatial variations in sterodynamic sea level, GRD effects, and GIA as well as small-scale VLM. All of the records show considerable inter-annual to decadal variability, with peak-to-peak fluctuations as large as 10 cm, and this variability is generally well reproduced by the BHM, albeit with a tendency to underestimate its magnitude. The degree of underestimation varies across locations and largely depends on the extent to which the local sea-level variability can be explained by the five leading EOFs over the Mediterranean basin. This is not an issue since most of the unexplained variability is captured by the measurement error term $(e_{i,l})$, thus ensuring consistency and inflating the uncertainty accordingly.

For comparison, we have also plotted estimates of sea-level changes from an EOF reconstruction at the four selected tide gauge sites (Figure 2). The EOF reconstruction is based on the same tide gauge data as the BHM and follows the formulation described by Calafat et al. (2014). At sites where VLM is small, the two reconstructions give similar time series of sea level, however at sites with strong VLM signals, such as Khalkis South (Figure 2c), the long-term changes are very different. This demonstrates an important point about EOF reconstructions. Because such reconstructions combine altimetry data with tide gauge observations without adequately accommodating the differences between GSL and RSL, it is unclear whether and to what extent the reconstructed long-term changes contain contributions from VLM. This is an important issue that has been addressed in our BHM.

We can form a more complete picture of the skill of the BHM in capturing the local sea-level variability by plotting the correlation between our estimates of detrended GSL changes and altimetry observations at each grid point (Figure 3a). In interpreting the correlation map, it is important to recall that the altimetry data are subsampled at every five grid points before being assimilated into the BHM while estimates are made on a $1/4^{\circ} \times 1/4^{\circ}$ grid (the full altimetry grid). This means that the correlations with altimetry data largely reflect model skill at unobserved locations. The average correlation over the whole Mediterranean basin is 0.83, indicating a very good agreement. However, there is significant spatial variation in the correlations, with values close to one in the Adriatic, Aegean, and Ionian Seas, and relatively low values (~0.6) in regions of high mesoscale activity (Fusco et al., 2003) such as the Alboran Sea, along the Algerian current, the Bonifacio Gyre, and along the Mid-Mediterranean Jet south of Crete. Mesoscale processes are characterized by small length scales (10–100 km), and so their effects on sea level are not well represented by the leading EOFs, nor can they be estimated with confidence at unobserved locations, which explains the lower correlations in the corresponding regions. Long-term sea-level changes are likely better captured by the BHM than the variability given that generally they are associated with larger length scales.

It is interesting to produce a similar correlation map but for the EOF reconstruction (Figure 3b). The regions of high and low correlation are the same as for the BHM, but, overall, the correlations are much lower, with an average value of 0.60. This is not surprising because the BHM assimilates both tide gauge observations and altimetry data, whereas the EOF reconstruction relies only on the tide gauge data to estimate the temporal amplitude of the altimetry-derived EOFs. In this regard, it is important to remark that the high correlations with the altimetry data in the case of the BHM (Figure 3a) are unlikely to be reflective of a similarly good performance over the pre-altimetry period. Looking now at the time series of sea level at an arbitrary point along the Algerian current (Figure 3c), we see that, unlike the BHM, the EOF reconstruction is, indeed, unable to reconstruct the sea-level variability in regions of high mesoscale activity. This can be interpreted as to mean that the existing tide gauge network does not provide enough information to capture the variability in such regions.





Figure 2. Annual inverse barometer (IB)-corrected relative sea level (RSL) as estimated by the Bayesian hierarchical model (BHM) (blue and black lines) along with estimates from an empirical orthogonal function (EOF) reconstruction (green line) and the observed RSL (blue line) from tide gauge records at (a) Marseille, (b) Trieste, (c) Khalkis South, and (d) Alexandria. We show the posterior mean for both the total RSL changes (blue line) and the long-term RSL changes due to sterodynamic changes, gravitational, rotational, and deformation (GRD) effects, glacial isostatic adjustment (GIA), and small-scale processes (black line). The gray shading represents ± 1 standard deviation (SD) associated with the estimate of long-term changes.

As an additional check of the BHM, we compare basin-average sea-level changes since 1960 as estimated by the BHM with those derived from the EOF reconstruction (Figure 3d). The two reconstructions agree very well on both the magnitude and the phase of the inter-annual variability, with a correlation of 0.92 for detrended time series. The long-term changes are also very similar in the two reconstructions, with an initial period of slow change from 1960 to 1990, followed by a period of rapid sea-level rise after 1990. Finally, we note that both reconstructions closely match the observed sea-level changes from altimetry for the period 1993–2018, giving us additional confidence in the skill of the BHM.

4. Results

We now investigate the BHM solutions with focus on long-term RSL changes. Hereafter, where we refer to total sea-level rates this includes the IB effect and excludes the contribution from small-scale processes (x_S) . We begin





Figure 3. (a) Correlation of detrended annual geocentric sea level (GSL) changes (inverse barometer (IB) effect removed) derived from the Bayesian hierarchical model (BHM) with observed changes from satellite altimetry at each grid point for the period 1993–2018. (b) The same as (a) but for estimates of sea-level changes obtained from an empirical orthogonal functions (EOF) reconstruction. (c) Observed (detrended) sea level changes (red line) at the point denoted by a black cross in panel (a) compared with estimates derived from the BHM (blue line) and an EOF reconstruction (green line). (d) Comparison of basin-average sea-level changes since 1960 as estimated by the BHM (blue and black lines) with those computed using an EOF reconstruction (green line). We show the posterior mean for both the total relative sea level (RSL) changes (blue line) and the long-term RSL changes due to sterodynamic changes, gravitational, rotational, and deformation (GRD) effects, glacial isostatic adjustment (GIA), and small-scale processes (black line). The gray shading represents ± 1 standard deviation (SD) associated with the estimate of long-term changes. The basin-average sea-level changes from altimetry are also shown (red line).

by analyzing time series of the basin-average rate of sea-level change for the period 1960–2018 (Figure 4a). The time series show that the total rate was relatively small and mostly negative before 1990, at which point it started to increase rapidly reaching a maximum value of 4.6 ± 0.6 mm yr⁻¹ (posterior mean ±1SD) in 2000 that has largely persisted to this day. The average RSL rate for the periods 1960–1989 and 2000–2018 was, respectively, -0.3 ± 0.5 mm yr⁻¹ and 3.6 ± 0.3 mm yr⁻¹ (Figure 4b). Our approach also provides estimates of the contributions from individual sources to the total RSL rate. For the period before 1990, we find that sterodynamic changes and GRD effects made small contributions of opposed sign that largely canceled one another out, with average rates of -0.2 ± 0.5 mm yr⁻¹ and 0.3 ± 0.2 mm yr⁻¹ (Figure 4b), respectively. The contribution from GIA, which is constant over the whole period, is relatively small with a positive value of 0.2 ± 0.1 mm yr⁻¹. For the period 2000–2018, sterodynamic changes and GRD effects both made positive and comparable contributions to the total rate of change, with values of 1.7 ± 0.3 mm yr⁻¹ and 1.5 ± 0.2 mm yr⁻¹ (Figure 4b), respectively. The contributions to the total rate of change, with values of 1.7 ± 0.3 mm yr⁻¹ and 1.5 ± 0.2 mm yr⁻¹.

We now turn our attention to the local rate of RSL change, focusing on the period 2000–2018 as this is when the highest rates occur. The spatial structure of RSL rates is characterized by a single-sign pattern of positive rates that displays considerable geographical variation (Figure 5a). The highest rates (>4 mm yr⁻¹) are found in the Tyrrhenian, Aegean, and Levantine Seas, whereas the lowest rates (<2 mm yr⁻¹) occur mainly along the Cretan



Journal of Geophysical Research: Oceans



Figure 4. (a) Time series of basin-average relative sea level (RSL) rate (black line) for the period 1960–2018 together with the contributions from sterodynamic sea level (red line), gravitational, rotational, and deformation (GRD) effects (blue line), glacial isostatic adjustment (GIA) (orange line), and inverse barometer (IB) effect (green line). The lines represent posterior means whereas the shading denotes ± 1 standard deviation (SD). (b) Posterior mean (central mark), interquartile range (shaded box), and 5%–95% credible interval (whiskers) for the basin-average RSL rate (gray) and the contributions from sterodynamic sea level (red), GRD effects (blue), and GIA (orange) averaged over the periods 1960–1989 and 2000–2018.

Passage. Hence, since 2000, RSL has been rising everywhere in the Mediterranean at a rate that varies significantly across space.

Focusing now on the individual contributions, we find that the spatial structure of the sterodynamic rates closely resembles that of the total RSL rates, with values that range from -0.1 to 3.1 mm yr^{-1} (Figure 5b). The GRD rates are positive everywhere (Figure 5c), with values ranging from 1 mm yr⁻¹ in the Northern Adriatic Sea and the Eastern Levantine Sea to >1.6 mm yr⁻¹ in the Alboran Sea and south of the Ionian Sea. The relatively large gradient in the GRD rates in the North Western Mediterranean and the Adriatic Sea is associated with melting of the Greenland ice sheet. Indeed, the zero-line in the Greenland sea-level fingerprint is situated over the North Sea, with a consequent sharp gradient in the Northern Mediterranean. The sharp gradient in the eastern sector of the Levantine Sea reflects land uplift due to groundwater depletion in the Middle East (Rodell et al., 2018; Wada et al., 2010). The GIA contribution is relatively small and displays a pattern of meridional variation wherein rates increase from small negative values (-0.2 mm yr^{-1}) along most of the African coast and in the easternmost part of the Levantine Sea to positive values of up to 0.5 mm yr⁻¹ in the Western Mediterranean (Figure 5d). The contribution from the IB effect is small and positive over most of the basin, with values ranging from -0.1 mm yr^{-1} in the Eastern Levantine Sea to just over 0.4 mm yr⁻¹ in the Central Mediterranean and the Aegean Sea (Figure 5e).

To provide a more quantitative description of the spatial structure of the rates, we have computed spatially averaged rates of RSL and those of the individual contributions over the five regions depicted in Figure 1 for the period 2000–2018 (Table 2). The Adriatic, Aegean, and Levantine Seas show the highest RSL rates, with spatially averaged values of 3.9 ± 0.5 mm yr⁻¹, 4.1 ± 0.4 mm yr⁻¹, and 3.9 ± 0.4 mm yr⁻¹, respectively. The lowest RSL rates are found in the central Mediterranean (3.5 ± 0.4 mm yr⁻¹). The difference in RSL rates between these regions is largely explained by spatial variations in the sterodynamic rates (Table 2), which likely reflect a steric effect associated with changes in the thermohaline circulation of the Mediterranean Sea (as we discuss in the next section). Indeed, the spatially averaged sterodynamic rate over the Levantine Sea is almost twice that over the central Mediterranean (2.5 ± 0.3 mm yr⁻¹ compared to 1.4 ± 0.3 mm yr⁻¹). The GRD rates are similar and positive across the five regions ($\sim 1.5 \pm 0.2$ mm yr⁻¹), whereas the GIA rates are relatively small (up to 0.3 mm yr⁻¹). The contribution from the IB effect is similar in magnitude to that from GIA, with a maximum value of 0.4 ± 0.2 mm yr⁻¹ in the Aegean Sea and a minimum value of 0.1 ± 0.2 mm yr⁻¹ in the Levantine Sea.



Journal of Geophysical Research: Oceans



Figure 5. Pattern of relative sea level (RSL) rates over the period 2000–2018 as estimated by the Bayesian hierarchical model (BHM) for (a) the total RSL (including the inverse barometer (IB) effect) and the contributions from (b) sterodynamic changes, (c) gravitational, rotational, and deformation (GRD) effects, and (d) glacial isostatic adjustment (GIA). The contribution from the IB effect (computed outside the BHM) is also shown in panel (e).

5. Conclusions and Discussion

Our analysis of data from tide gauges and altimetry has shown that spatially averaged sea level over the Mediterranean Sea fell at a rate of -0.3 mm yr^{-1} between 1960 and 1989 (Figure 4), in agreement with findings from past studies (Marcos & Tsimplis, 2008; Tsimplis & Baker, 2000; Tsimplis et al., 2005). This contrasts with a rate of global mean sea-level rise of 0.9 mm yr⁻¹ in the same period (based on data from Frederikse et al., 2020). The fall in sea level during 1960–1989 was largely due to a positive anomaly in atmospheric pressure over the Mediterranean pushing water out of the basin, which previous studies have linked to the NAO (Gomis et al., 2008;

Table 2

Spatially Averaged Rate of RSL Change (Posterior Mean \pm SD) and the Individual Contributions for the Period 2000–2018 Over the Five Subregions Denoted in Figure 1

Subregion	Total (mm yr ⁻¹)	Sterodynamic (mm yr ⁻¹)	GRD (mm yr ⁻¹)	GIA (mm yr ⁻¹)	IB (mm yr ⁻¹)
Western Med.	3.6 ± 0.4	1.5 ± 0.3	1.5 ± 0.2	0.3 ± 0.1	0.2 ± 0.3
Adriatic Sea	3.9 ± 0.5	1.9 ± 0.3	1.4 ± 0.2	0.3 ± 0.2	0.3 ± 0.3
Central Med.	3.5 ± 0.4	1.4 ± 0.3	1.6 ± 0.2	0.2 ± 0.1	0.3 ± 0.2
Aegean Sea	4.1 ± 0.4	2.0 ± 0.4	1.5 ± 0.2	0.2 ± 0.1	0.4 ± 0.2
Levantine Sea	3.9 ± 0.4	2.5 ± 0.3	1.4 ± 0.2	0.0 ± 0.1	0.1 ± 0.2





Figure 6. Local deviations of the rate of sterodynamic sea-level change from the basin-average rate for the period 2000–2018 as estimated by the Bayesian hierarchical model (BHM).

Tsimplis et al., 2005; Tsimplis & Josey, 2001). However, even after accounting for the effect of this increase in atmospheric pressure, the rate of sea-level change in the Mediterranean $(0.4 \pm 0.5 \text{ mm yr}^{-1})$ over that period was still less than half the global average rate (0.9 mm yr^{-1}) . The relatively low rate of IB-corrected sea-level rise in the Mediterranean in the period 1960–1989 has been noticed by previous studies (e.g., Marcos & Tsimplis, 2008; Tsimplis & Baker, 2000), but the reasons for this remained unclear. Here we have shown that the low rate was due to the small and opposing contributions from sterodynamic changes and GRD effects. We have also found that, since 2000, Mediterranean sea level has been rising at an increased rate of $3.6 \pm 0.3 \text{ mm yr}^{-1}$, which is in line with the rate of global mean sea level rise for that period. This increased rate of rise has been driven by similar contributions from sterodynamic changes and GRD effects.

We have also shown that sea level changes in the Mediterranean Sea are highly non-uniform, with local rates that can deviate from the basin-mean rate by as much as 2 mm yr⁻¹ for the period 2000–2018. This spatial non-uniformity is largely controlled by changes in sterodynamic sea level and is particularly remarkable in the Eastern Mediterranean where the sterodynamic contribution is characterized by rapid sea-level rise in the Levantine and Aegean Seas and significantly slower rise in the Ionian Sea and along the Cretan passage. This trend pattern emerges more clearly when plotting deviations from the basin-mean sterodynamic rate (Figure 6). The pattern of deviations shows a striking contrast between negative deviations in the central Mediterranean and positive ones in all the adjacent seas (Adriatic, Aegean, and Levantine). This pattern is highly consistent with the changes in the thermohaline circulation and sea level that followed the end of the Eastern Mediterranean Transient (EMT) in 1995 (Calafat, Jordà, et al., 2012; Roether et al., 1996). The EMT started in 1987 and was characterized by a temporary shift in the primary sites of dense water formation in the Eastern Mediterranean from the Adriatic Sea to the Aegean Sea. This shift ended in 1995 when the Aegean stopped exporting significant amounts of dense water and the Adriatic Sea became again the primary deep-water source (Theocharis et al., 2002). These post-1995 changes in water density and circulation had a strong signature in sea level marked by a sea-level fall in the Ionian Sea and a rise in the Aegean and Levantine Seas (Calafat, Jordà, et al., 2012). The pattern of sterodynamic trends for the period 2000-2018 (Figure 6) likely reflects some of the post-EMT changes, although confirming this would require a detailed hydrographic analysis, which is beyond the scope of this paper.

Our new approach provides coastal planners with a valuable tool to obtain robust and up-to-date estimates of local sea-level changes, while distinguishing long-term changes from short-term variability and informing about the causes of the changes. This will allow planners to make more confident and timely decisions regarding coastal adaptation for sea-level rise in the Mediterranean Sea. Additionally, since our new estimates are only based on observations, they can be used to evaluate simulated sea-level from climate models, on which sea-level projections are based, thus contributing to efforts to reduce model uncertainty and, in turn, further benefiting coastal planning decisions.

Data Availability Statement

The Bayesian solutions as well as the input data needed to run the BHM and replicate the results shown in this paper are available at Zenodo via https://doi.org/10.5281/zenodo.7046702 (Calafat et al., 2022). The code that implements the BHM is also available at Zenodo via https://doi.org/10.5281/zenodo.6797648 (Calafat, 2022).

References

Banerjee, S., Carlin, B. P., & Gelfand, A. E. (2004). Hierarchical modeling and analysis for spatial data. Chapman and Hall.

Calafat, F. M. (2022). Code supporting: "The Sources of Sea-Level Changes in the Mediterranean Sea since 1960" (1.0). Zenodo. https://doi. org/10.5281/zenodo.6797648

Calafat, F. M., Chambers, D. P., & Tsimplis, M. N. (2012). Mechanisms of decadal sea level variability in the eastern North Atlantic and the Mediterranean Sea. Journal of Geophysical Research, 117, C09022. https://doi.org/10.1029/2012JC008285

Calafat, F. M., Chambers, D. P., & Tsimplis, M. N. (2014). On the ability of global sea level reconstructions to determine trends and variability. Journal of Geophysical Research: Oceans, 119, 1572–1592. https://doi.org/10.1002/2013JC009298

Calafat, F. M., Frederikse, T., & Horsburgh, K. (2022). Reconstruction of Mediterranean sea-level changes and contributions for 1960–2018 (2.3) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.7046702

Acknowledgments

We acknowledge the PSMSL for the tide gauge data and CMEMS for the altimetry data. F.M.C. and K.H. both were supported by the EuroSea project funded by the European Union's Horizon 2020 research and innovation programme under grant agreement no. 862626.

- Calafat, F. M., & Gomis, G. (2009). Reconstruction of Mediterranean sea level fields for the period 1945–2000. *Global and Planetary Change*, 66, 225–234. https://doi.org/10.1016/j.gloplacha.2008.12.015
- Calafat, F. M., & Jordà, G. (2011). A Mediterranean sea level reconstruction (1950–2008) with error budget estimates. *Global and Planetary Change*, 79, 118–133. https://doi.org/10.1016/j.gloplacha.2011.09.003
- Calafat, F. M., Jordà, G., Marcos, M., & Gomis, D. (2012). Comparison of Mediterranean sea level variability as given by three baroclinic models. Journal of Geophysical Research, 117, C02009. https://doi.org/10.1029/2011JC007277
- Carpenter, B., Gelman, A., Hoffman, M. D., Lee, D., Goodrich, B., Betancourt, M., et al. (2017). Stan: A probabilistic programming language. *Journal of Statistical Software*, 76, 1–32. https://doi.org/10.18637/jss.v076.i01
- Carrère, L., & Lyard, F. (2003). Modeling the barotropic response of the global ocean to atmospheric wind and pressure forcing Comparisons with observations. *Geophysical Research Letters*, 30, 1275. https://doi.org/10.1029/2002GL016473
- Clark, J. A., & Lingle, C. S. (1977). Future sea-level changes due to West Antarctic ice sheet fluctuations. *Nature*, 269, 206–209. https://doi.org/10.1038/269206a0
- Dangendorf, S., Hay, C., Calafat, F. M., Marcos, M., Piecuch, C. G., Berk, K., & Jensen, J. (2019). Persistent acceleration in global sea-level rise since the 1960s. *Nature Climate Change*, 9, 705–710. https://doi.org/10.1038/s41558-019-0531-8
- Dangendorf, S., Marcos, M., Wöppelmann, G., Conrad, C. P., Frederikse, T., & Riva, R. (2017). Reassessment of 20th century global mean sea level rise. Proceedings of the National Academy of Sciences of the United States of America, 114, 5946–5951. https://doi.org/10.1073/ pnas.1616007114
- Datta, A., Banerjee, S., Finley, A. O., & Gelfand, A. E. (2016). Hierarchical nearest-neighbor Gaussian process models for large geostatistical datasets. Journal of the American Statistical Association, 111, 800–812. https://doi.org/10.1080/01621459.2015.1044091
- Frederikse, T., Landerer, F., Caron, L., Adhikari, S., Parkes, D., Humphrey, V. W., et al. (2020). The causes of sea-level rise since 1900. Nature, 584(7821), 393–397. https://doi.org/10.1038/s41586-020-2591-3
- Fusco, G., Manzella, G. M. R., Cruzado, A., Gačić, M., Gasparini, G. P., Kovačević, V., et al. (2003). Variability of mesoscale features in the Mediterranean Sea from XBT data analysis. Annales Geophysicae, 21, 21–32. https://doi.org/10.5194/angeo-21-21-2003
- Gomis, D., Ruiz, S., Sotillo, M. G., Álvarez-Fanjul, E., & Terradas, J. (2008). Low frequency Mediterranean Sea level variability: The contribution of atmospheric pressure and wind. *Global and Planetary Change*, 63 (2–3), 215–229. https://doi.org/10.1016/j.gloplacha.2008.06.005
- Hay, C. C., Morrow, E., Kopp, R. E., & Mitrovica, J. X. (2015). Probabilistic reanalysis of twentieth-century sea-level rise. *Nature*, *517*, 481–484. https://doi.org/10.1038/nature14093
- Holgate, S. J., Matthews, A., Woodworth, P. L., Rickards, L. J., Tamisiea, M. E., Bradshaw, E., et al. (2013). New data systems and products at the Permanent Service For Mean Sea Level. *Journal of Coastal Research*, 29(3), 493–504. https://doi.org/10.2112/JCOASTRES-D-12-00175.1
- Kalnay, E., Kanamitsu, M., Kirtler, R., Collins, W., Deaven, D., Gandin, L., et al. (1996). The NCEP/NCAR 40-year reanalysis project. Bulletin of the American Meteorological Society, 77, 437–471. https://doi.org/10.1175/1520-0477(1996)077<0437:tnyrp>2.0.co;2
- Marcos, M., & Tsimplis, M. N. (2008). Coastal sea level trends in Southern Europe. Geophysical Journal International, 175(1), 70–82. https:// doi.org/10.1111/j.1365-246X.2008.03892.x
- Meyssignac, B., Calafat, F. M., Somot, S., Rupolo, V., Stocchi, P., Llovel, W., & Cazenave, A. (2011). Two-dimensional reconstruction of the Mediterranean sea level over 1970–2006 from tide gauge data and regional ocean circulation model outputs. *Global and Planetary Change*, 77, 49–61. https://doi.org/10.1016/j.gloplacha.2011.03.002
- Mohamed, B., Abdallah, A. M., Alam El-Din, K., Nagy, H., & Shaltout, M. (2019). Inter-annual variability and trends of sea level and sea surface temperature in the Mediterranean sea over the last 25 years. Pure and Applied Geophysics, 176, 3787–3810. https://doi.org/10.1007/ s00024-019-02156-w
- Orlić, M., Pasarić, M., & Pasarić, Z. (2019). Mediterranean sea-level variability in the second half of the twentieth century: A Bayesian approach to closing the budget. In I. Vilibić, K. Horvath, & J. Palau (Eds.), *Meteorology and climatology of the Mediterranean and black seas. Pageoph* topical volumes. Birkhäuser. https://doi.org/10.1007/978-3-030-11958-4_15
- Piecuch, C. G., Huybers, P., & Tingley, M. P. (2017). Comparison of full and empirical Bayes approaches for inferring sea-level changes from tide-gauge data. *Journal of Geophysical Research: Oceans*, 122, 2243–2258. https://doi.org/10.1002/2016JC012506
- Rasmussen, C. E., & Williams, C. K. I. (2006). Gaussian processes for machine learning. MIT Press.
- Reimann, L., Vafeidis, A. T., Brown, S., Hinkel, J., & Tol, R. S. J. (2018). Mediterranean UNESCO World Heritage at risk from coastal flooding and erosion due to sea-level rise. *Nature Communications*, 9, 4161. https://doi.org/10.1038/s41467-018-06645-9
- Richter, K., Riva, R. E. M., & Drange, H. (2013). Impact of self-attraction and loading effects induced by shelf mass loading on projected regional sea level rise. *Geophysical Research Letters*, 40(6), 1144–1148. https://doi.org/10.1002/grl.50265
- Rodell, M., Famiglietti, J. S., Wiese, D. N., Reager, J. T., Beaudoing, H. K., Landerer, F. W., & Lo, M.-H. (2018). Emerging trends in global freshwater availability. *Nature*, 557, 651–659. https://doi.org/10.1038/s41586-018-0123-1
- Roether, W., Manca, B. B., Klein, B., Bregant, D., Georgopoulos, D., Beitzel, et al. (1996). Recent changes in eastern Mediterranean deep waters. Science, 271(5247), 333–335. https://doi.org/10.1126/science.271.5247.333
- Tamisiea, M. E. (2011). Ongoing glacial isostatic contributions to observations of sea level change. Geophysical Journal International, 186(3), 1036–1044. https://doi.org/10.1111/j.1365-246X.2011.05116.x
- Theocharis, A., Klein, B., Nittis, K., & Roether, W. (2002). Evolution and status of the eastern Mediterranean transient (1997–1999). Journal of Marine Systems, 33(34), 91–116. https://doi.org/10.1016/S0924-7963(02)00054-4
- Tsimplis, M. N., Álvarez-Fanjul, E., Gomis, D., Fenoglio-Marc, L., & Pérez, B. (2005). Mediterranean Sea level trends: Atmospheric pressure and wind contribution. *Geophysical Research Letters*, 32, L20602. https://doi.org/10.1029/2005GL023867
- Tsimplis, M. N., & Baker, T. F. (2000). Sea level drop in the Mediterranean sea: An indicator of deep water salinity and temperature changes? Geophysical Research Letters, 27(12), 1731–1734. https://doi.org/10.1029/1999GL007004
- Tsimplis, M. N., & Josey, S. (2001). Forcing the Mediterranean sea by atmospheric oscillations over the North Atlantic. *Geophysiscal Research Letters*, 28(5), 803–806. https://doi.org/10.1029/2000GL012098
- Wada, Y., van Beek, L. P. H., van Kempen, C. M., Reckman, J. W. T. M., Vasak, S., & Bierkens, M. F. P. (2010). Global depletion of groundwater resources. *Geophysical Research Letters*, 37, L20402. https://doi.org/10.1029/2010GL044571